Author's response to reviews

Title: Glucocorticoids with different chemical structures but similar glucocorticoid receptor potency regulate subsets of common and unique genes in human trabecular meshwork cells.

Authors:

Alissar Nehme (nehme_alissar@allergan.com)
Edward K Lobenhofer (ed@lobenhofer.com)
W Daniel Stamer (dstamer@eyes.arizona.edu)
Jeffrey L Edelman (edelman_jeffrey@allergan.com)

Version: 5 Date: 4 September 2009

Author's response to reviews: see over
Dear Dr Norton,

**RE: Resubmission of revised research article 2193986932560370**

On behalf of my co-authors, I am pleased to submit to *BMC Medical Genomics*, our revised manuscript entitled:

*Glucocorticoids with different chemical structures but similar glucocorticoid receptor potency regulate subsets of common and unique genes in human trabecular meshwork cells*

My co-authors and I would like to thank you again for accepting our manuscript for publication in *BMC Medical Genomics* (your email dated 17th August 2009). We have made two minor changes to text within the revised manuscript (changes are indicated in red font).

Sincerely,

Alissar Nehme, PhD